Author's response to reviews

Title: Diagnostic value of Diffusion-Weighted MR imaging in thyroid disease: application in differentiating benign from malignant disease

Authors:

yingwei wu (wuyw0103@hotmail.com)
xiuhui yue (wuyw0103@gmail.com)
weiwen shen (sww_281@sina.com)
Yushan Du (laifa1015@163.com)
Xiaofeng Tao (cjr.taokiaofeng@vip.163.com)
Ying Yuan (yuany83@163.com)
Cheuk Ying Tang (cheukying.tang@mssm.edu)

Version: 2 Date: 11 November 2012

Author's response to reviews:

We have already stated the ethics statement in the methods section. Authors contribution and Competing Interest section was also added to the main manuscript.